Author's response to reviews

Title: Diagnosis of Noonan syndrome and related disorders using Target Next Generation Sequencing.

Authors:

Francesca R Lepri (francescaromana.lepri@opbg.net)
Rossana Scavelli (rscavelli@illumin.com)
Maria Cristina Digilio (mcdigilio@opbg.net)
Maria Gnazzo (maria.gnazzo@opbg.net)
Simona Grotta (simona.grotta@opbg.net)
Maria Lisa Dentici (marialisa.dentici@opbg.net)
Elisa Pisaneschi (elisa.pisaneschi@opbg.net)
Pietro Sirleto (pietro.sirleto@opbg.net)
Rossella Capolino (rossella.capolino@opbg.net)
Anwar Baban (anwar.baban@opbg.net)
Serena Russo (serena.russo@opbg.net)
Tiziana Franchin (tiziana.franchin@opbg.net)
Adriano Angioni (adriano.angioni@opbg.net)
Bruno Dallapiccola (bruno.dallapiccola@opbg.net)

Version: 2 Date: 8 October 2013

Author's response to reviews:

In the Method section on page 4 I include this comment about ethics statement:
The study was approved by the institutional scientific board of Bambino Gesù Children Hospital and was conducted in accordance with the Helsinki Declaration.